Author's response to reviews

Title: Motives for (not) participating in a lifestyle intervention trial

Authors:

Jeroen Lakerveld (j.lakerveld@vumc.nl)
Wilhelmina IJzelenber (helma.ijzelenberg@vumc.nl)
Maurits W van Tulder (mw.vantulder@vumc.nl)
Irene M Hellemans (i.hellemans@vumc.nl)
Jan A Rauwerda (ja.rauwerda@vumc.nl)
Albert C van Rossum (ac.vrossum@vumc.nl)
Jaap C Seidell (jaap.seidell@falw.vu.nl)

Version: 3 Date: 29 March 2008

Author's response to reviews:

Dear Iratxe Puebla

Please find enclosed the second revised version of the manuscript "Motives for (not) participating in a lifestyle intervention trial".

We changed the legend of Figure 2 according to the suggestion of reviewer 3.

As suggested, we went through the manuscript formatting checklist one more time and made several improvements to the lay-out.

We also considered the contributing role of Dr Seidell to the manuscript, and changed the authors' contribution section accordingly.

Both figures are now cropped as closely as possible to minimise white space around the image.

We hope the changes further improved the manuscript.

Sincerely,

Jeroen Lakerveld
Corresponding author